Author's response to reviews

Title: Identifying Quality Improvement intervention publications - A comparison of electronic search strategies

Authors:

- Susanne Hempel (susanne_hempel@rand.org)
- Lisa Rubenstein (lisa_rubenstein@rand.org)
- Roberta Shanman (Roberta_shanman@rand.org)
- Robbie Foy (R.Foy@leeds.ac.uk)
- Su Golder (su.golder@york.ac.uk)
- Danz Marjorie (mjsdanz@gmail.com)
- Shekelle Paul (paul_shekelle@rand.org)

Version: 3 Date: 20 June 2011

Author's response to reviews:

Dear Dr. Sales,

I've made the two changes mentioned by the reviewer (add the reference placeholder, add the database and search period) - both on page 11 in the manuscript and highlighted in yellow.

Many thanks, Susanne.